Author’s response to reviews

Title: Effect of Iron Deficiency Anemia on HbA1c in Diabetic Patients at Tikur Anbessa Specialized Teaching Hospital, Addis Ababa Ethiopia

Authors:

Absra Solomon (absrasolomon66@gmail.com)
Mintewab Hussein (mintewabhussein@gmail.com)
Mikias Negash (mikiasn2@gmail.com)
Abdurezak Ahmed (abdurezakahmed@gmail.com)
Fitsum Bekele (fbt1019@yahoo.com)
Daniel Kahase (danielasalfew@gmail.com)

Version: 2 Date: 13 Jul 2018

Author’s response to reviews:

Rebuttal: point by point response to referees’ comments

Effect of Iron Deficiency Anemia on HbA1c in Diabetic Patients at Tikur Anbessa Specialized Teaching Hospital, Addis Ababa Ethiopia

Reviewer reports:

Alauldeen Mudhafar Zubair, M.D. (Reviewer 1): Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated versions of manuscripts, graphs, supporting materials or other aspects of your report which cannot be included in a text format.

Please overwrite this text when adding your comments to the authors.

Thank you Dear Alauldeen Mudhafar Zubair for the comments and guiding.

* We have done accordingly
Karen Cordovil, M.E. (Reviewer 2): The authors,

In the article summary you mention that - Socio-demographic and clinical conditions were collected through a structured questionnaire - but there is nothing reported in the methodology, please describe how it was done in detail.

Thank you Dear Karen Cordovil, we appreciate your effort.

We have done accordingly------ (Method section line 1 page 5)

Fitsum Bekele and Absra Solomon